Author's response to reviews

Title: Physical activity, sedentary behaviors, and estimated insulin sensitivity and secretion in pregnant and non-pregnant women

Authors:

Anna Gradmark (anna.gradmark@medicin.umu.se)
Jeremy Pomeroy (pomeroyj@niddk.nih.gov)
Frida Renström (frenstro@hsph.harvard.edu)
Susanne Steiginga (s_steiginga@hotmail.com)
Margareta Persson (margareta.persson@nurs.umu.se)
Antony Wright (Antony.Wright@mrc-epid.cam.ac.uk)
Les Bluck (Les.Bluck@mrc-hnr.cam.ac.uk)
Magnus Domellöf (magnus.domellof@pediatri.umu.se)
Steven E Kahn (skahn@u.washington.edu)
Ingrid Mogren (ingrid.mogren@obgyn.umu.se)
Paul W Franks (paul.franks@med.lu.se)

Version: 7 Date: 16 June 2011

Author's response to reviews:

The purpose of our revisions are to fix minor typographical errors in our manuscript. Here is a list of the changes we have made:

1. In Abstract and Results section added the word “regression” before r2.
2. In Abstract, Methods, second row: “nonpregnant” changed to “non-pregnant”.
3. In the next section Blood analyses, added the words “the challenge” before “and at 30, 60 and 120 minutes”.
4. In the section Estimates of insulin sensitivity and B-cell response the words “oral glucose tolerance test” abbreviated to “OGTT”.
5. In Background, page 4, “physical activity and glucose homeostasis”, changed to “physical activity on glucose homeostasis”.
6. In Background, page 5, row 5: (PAEE) inserted after “physical activity energy expenditure”.
7. In Background, page 5, second para, row 2: “intensity” should be inserted after “moderate-to-vigorous”.
8. In Methods, Energy expenditure, row 3: ’10 days’ changed to 'ten days'.
10. In Methods, Blood analysis, row 1: ‘clinical research center’ changed to ‘Clinical Research Center’.
glucose tolerance test’ changed to “OGTT”.
12. In Results, row 2: 'doubly labeled water' replaced by 'DLW'.
13. In Discussion, 4th para, row 4: 'B-cell' inserted before 'response'.
15. Under the section Estimates of insulin sensitivity and #-cell response, added a closing bracket for this equation: [1,194 + (4.724*Ins0) – (117.0*Glu60) + (1.414*Ins60)].
16. In the same section, for consistency, the equation for the oral disposition index modified as follows: [(#Ins 0-30 / #Glu 0-30) x 1/fasting insulin].